Author's response to reviews

Title: Extremely short duration high intensity training substantially improves insulin action in young healthy males

Authors:
John Babraj (j.babraj@hw.ac.uk)
Niels Vollaard (n.vollaard@hw.ac.uk)
C Keast (ck51@hw.ac.uk)
F M Guppy (fmg1@hw.ac.uk)
G Cottrell (gc29@hw.ac.uk)
James A Timmons (jamie.timmons@gmail.com)

Version: 3 Date: 13 January 2009

Author's response to reviews:
Final editing adjustments made.